Author’s response to reviews

Title: Caregiver exposure to critical events and distress in home-based palliative care in Germany A cross-sectional study using the Stressful Caregiving Adult Reactions to Experiences of Dying (SCARED) Scale

Authors:

Michael Galatsch (michael.galatsch@uni-wh.de)
Holly Prigerson (hgp2001@med.cornell.edu)
Wilfried Schnepp (schnepp@uni-wh.de)
Friederike Sayn-Wittgenstein-Hohenstein (f.wittgenstein@hs-osnabrueck.de)
Jian Li (Jian.Li@uni-duesseldorf.de)

Version: 2 Date: 09 Aug 2018

Author’s response to reviews:

Dear Editors,

We thank you for the comments which are very constructive and helpful, regarding our manuscript of “Caregiver exposure to critical events and distress in home-based palliative care in Germany A cross-sectional study using the Stressful Caregiving Adult Reactions to Experiences of Dying (SCARED) Scale.” PCAR-D-18-00073R1

We have revised the manuscript according to your comments. In addition, a scientist whose native language is English reviewed the final version again (all revised parts are in red).

We hope that this revised manuscript is now suitable for publication. Should there be additional queries do not hesitate to contact us immediately.

Thank you very much!

Sincerely yours,

Michael Galatsch